Author's response to reviews

Title: Obesity, metabolic syndrome, impaired fasting glucose and microvascular dysfunction: a principal component analysis approach

Authors:

Diogo G Panazzolo (guarnipana@yahoo.com.br)
Fernando L Sicuro (fsicuro@gmail.com)
Ruth Clapauch (rclapauch@uol.com.br)
Priscila A Maranhão (priscilamaranhao@gmail.com)
Eliete Bouskela (eliete_bouskela@yahoo.com.br)
Luiz G Kraemer-Aguiar (gkraemer@iq.com.br)

Version: 4 Date: 16 October 2012

Author's response to reviews:


Professor Jeetesh Patel.

Associate Editor

BMC Cardiovascular Disease

Dear Prof. Patel

Please find enclosed the revised manuscript “Obesity, metabolic syndrome, impaired fasting glucose and microvascular dysfunction: a principal component analysis approach”.

First of all, we would like to thank you for the opportunity to resubmit and also to thank the reviewers for comments and constructive criticisms. The language has been revised accordingly to your suggestion at www.edanzediting.com/bmc1.

The final version is on “Clear Format” and the copy edited version is on “Tracked Format”.

Sincerely yours,

Luiz Guilherme Kraemer-Aguiar

Associate Professor

State University of Rio de Janeiro